Author's response to reviews

Title: Visual Neglect in Posterior Cortical Atrophy

Authors:

Katia Andrade (andrade-k@hotmail.com)
Dalila Samri (dalila.samri@gmail.com)
Marie Sarazin (marie.sarazin@psl.aphp.fr)
Leonardo C Souza (leocruzsouza@hotmail.com)
Laurent Cohen (laurent.cohen@psl.aphp.fr)
Michel Thiebaut de Schotten (michel.thiebaut@gmail.com)
Bruno Dubois (bruno.dubois@psl.aphp.fr)
Paolo Bartolomeo (paolo.bartolomeo@upmc.fr)

Version: 4 Date: 8 August 2010

Author's response to reviews: see over
Paris, 8th August 2010

Dear Editor,

Thank you for accepting our manuscript entitled “Visual neglect in posterior cortical atrophy” (by Andrade, Samri, Sarazin, Souza, Cohen, Thiebaut de Schotten, Dubois and Bartolomeo) for publication in the BMC Neurology journal. We are very happy and grateful.

In this final version, we accomplished all the formatting changes requested, including both the major and the minor revisions. Below is a point-by-point description of the changes made.

We look forward to hearing from you.

Sincerely yours,
Katia Andrade, MD, PhD student
Paolo Bartolomeo, MD, PhD